Author's response to reviews

Title: Efficacy and safety of mycophenolate mofetil treatment in IgA nephropathy: a systematic review

Authors:

Youyuan Chen (bigyouyuan@163.com)
Yumin Li (li_ymbnl@sina.com)
Shenglin Yang (Yang_SL_s@126.com)
Yan Li (liysum@126.com)
Min Liang (liangmnfyy@126.com)

Version: 4 Date: 12 November 2014

Author's response to reviews: see over
Dear editors:

Firstly, on behalf of all the authors, I will thank you for your recognition to our work and your helpful advices for revising the manuscript.

Here comes the point-by-point response to your concerns.

1. Please remove the copies of the revised manuscript and the revision with track changes from the additional files to comply with BMC guidelines (http://www.biomedcentral.com/bmcnephrol/authors/instructions/researcharticle#preparing-additional-files).

   Thank you for your advices. We have removed the copies of the revised manuscript and the revision with track changes from the additional files and strictly complied with the BMC guidelines.

2. Copyediting: We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

   Thank you for your advices. We have used a professional language editing service from Edanz for copyediting our manuscript.

3. Formatting - You now have an opportunity to check that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

   Thank you for your advices. We have checked our revised manuscript conforming to the journal style as the website you provided.